Author’s response to reviews

Title: Etiology of right ventricular restrictive physiology early after repair of Tetralogy of Fallot in pediatric patients

Authors:

Sandeep Bhushan (dr.sandeep07@yahoo.com)
Xin Huang (604680675@qq.com)
Fan Xu (xufanambition@126.com)
Pengxiao Su (365098438@qq.com)
Ting Wang (463593202@qq.com)
Xiaoke Sun (273310417@qq.com)

Version: 1 Date: 17 Apr 2019

Author’s response to reviews:

1. Grammatical errors and misspelling are corrected.
2. Most of numbers in result past has been removed.
3. The contest of conclusion which was almost identical to that of results had been changed.
4. Some new insights variables are added to make this manuscript better, especially data obtained for echocardiography such as TAPSE, PASP and TAPSE/PASP ratio.
5. Results part and discussion part have been corrected.